
• Cookies
○ 24.57mA/hr:  Average Stacking rate for Week of Mar30-Apr5

•

A slight increase of 0.88mA/hr

• Weekly pbar production
• From management

○ Safety
○ Documentation - will be updating.

2009-04-16 Pbar Department Meeting Thursday, April 16, 2009
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Cryo & TeV Problems

Mar30 through Apr 5 Ave Stack Rate Record

• Last Week
• Prior to Debuncher

• I:V714 – still tripping -

□ Has not tripped in 3-4 days.

• AP1 correctors not following reference

• Lithium breach into Lens’ water -- replace

• Lens running at reduced gradient
○ 2400V – 2600V – failure – 2200V – 2300V 

• Spares
□ 10mm-8 finished June 8th.

□ 10mm-1 recover transformer - in progress

□ 10mm-2 (water pipe developed leak away from transformer - tie rod nut missing 
also) and 10mm-3 (came out when ground current high) recovery planned

• Sweeping system PBKICK failure and “upgrades”

• Q731 failure

• Debuncher

• D:ISEPV trips on vacuum

• TWT VME Front End problems and TWT problems

□ Worth checking helix voltages

• D/A

• D:EKIK cable re-termination, 465 card

• D:EKIK module timing and OAC parameters

• Accumulator

• A:IKIK at 57KV, soon 58KV (other two are 61KV).

• Ground current outside of the machine/power supply

• Stacktail Equalizer tweak - hasn't helped at best.  May be hurting us a bit.

• Second 2-4GHz core TWT - CPTW02

• Transverse gain adjustment

□ Band 3 for both H and V.

• TWT trips

• Transfers

• Efficiency -

• I:HVF12 - more stable once voltage regulation turned back on.  Things got better yet again 
recently for no apparent reason.

• ARF4 curve frequency - taking first one at CF, then incrementing at 0.5Hz.  This is mostly 
for when we take a larger number of transfers.
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for when we take a larger number of transfers.

• General

• Water leaks fixed and more found

• Jobs

• Replace IB7 (when it fails)

• Convert Stacktail TWT#1 to second 2-4GHz Core cooling TWT

• Stacktail equalizer tweak

• Upstairs no beam for 4hrs

• Braze D:6Q11 water leak

• Next opportunity with Tech Division

• A4S12 is also leaking

• Valeri Proposals

• Re-introduce equalizers to core transverse systems  -on hold

• Re-install AP2 momentum collimator

• Need to request jaw modifications

• Move 4-8GHz kicker tank towards AP50 Pit

• Add second 4-8GHz kicker tank

• Bake 2 sectors

• New water manifolds for HV100 magnets

• Not doing HV102 and V105 manifolds for now

• Target Station

• New target SEM

• Water cooled Target Prep Work

• Fix the service elevators

• FESS maintenance

• Transport sump pump

• Compressor in AP10 and AP0

• Drive belts on unit for DRF1 racks

• Tunnel exhaust fans

• Shutdown

•

   Meetings Page 3    



•

•

•

   Meetings Page 4    



   Meetings Page 5    



   Meetings Page 6    



   Meetings Page 7    


